Author’s response to reviews

Title: A novel scoring system for prognostic prediction in d-galactosamine/lipopolysaccharide-induced fulminant hepatic failure BALB/c mice

Authors:

Bo Feng (xyfyfb_1@sina.com)
Sheng Ming Wu (wsm@proteomics.cn)
Sa Lv (lvsa@sina.com)
Feng Liu (liu1116m@sina.com)
Hong Song Chen (chen2999@sohu.com)
Yan Gao (gaoyan6384@163.com)
Fang Ting Dong (dft@proteomics.cn)
Lai Wei (weilai@pkuph.edu.cn)

Version: 4 Date: 18 December 2009

Author’s response to reviews: see over
Dear Editors:

We do appreciate for your instruction and advices. We have made the requested formatting changes and checked my manuscript for errors. As for ethical approval, we noted that animal studies were carried out in accordance with the Chinese National Research Council guidelines and approved by the Subcommittee on Research Animal Care and Laboratory Animal Resources of the Peking University People’s Hospital in Methods section. Please check it. If you still have some questions on revision, please contact us without any hesitation.

Thank you again and Best wishes

Sincerely yours

Lai WEI. Prof. MD,Ph.D
Bo FENG. Ph.D
Peking University People’s Hospital
Peking University Hepatology Institute